Author's response to reviews

Title: Upper gastrointestinal endoscopy for dyspepsia: Exploratory study of factors influencing patient compliance in Greece

Authors:

Eirini Oikonomidou (roikonomidou@yahoo.gr)
Foteini Anastasiou (fanast@hotmail.com)
Ioannis Pilipidis (ipilpidis@yahoo.gr)
Elias Kouroumalis (kouroum@med.uoc.gr)
Christos Lionis (lionis@galinos.med.uoc.gr)

Version: 4 Date: 18 January 2011

Author's response to reviews: see over
16 January 2010

Dear Editor,

On behalf of all the co-authors, I would like to thank you and the reviewer for the additional information. All your constructive comments were closely followed and an extensive linguistic editing was performed. We have revised our article and it is a pleasure for us providing you our point by point response to their comments.

Yours sincerely,

Eirini Oikonomidou

**Reviewer 1:**

1) **Response:** Thank you for the comment. We have rephrased the conclusion of the abstract in order to improve clarity.

2) **Response:** We have changed the labels of table 1

3) **Quality of written English:** We have reedited the article with the assistance of native English speaker